Author's response to reviews

Title: The association between demographic factors, health service factors and user satisfaction: results from a study from three casualty clinics in Norway

Authors:

Kirsten Danielsen (kirsten.danielsen@kunnskapssenteret.no)
Oyvind Andresen Bjertnaes (OyvindAndresen.Bjertnaes@kunnskapssenteret.no)
Andrew Garratt (Andrew.Garratt@kunnskapssenteret.no)
Oddvar Forland (Oddvar.Forland@uni.no)
Hilde Hestad Iversen (Hilde.Iversen@kunnskapssenteret.no)
Steinar Hunskaar (steinar.hunskar@isf.uib.no)

Version: 2 Date: 16 March 2010

Author's response to reviews:

An amendment is made to reference 15 in the reference list.